\begin{table}%t2
\par
\caption {\label{t:gcnocmd}Additional targets  grouped according to their location within the same-exposure field (see Sect.~2).}
  %\centerline
 {\scriptsize\begin{tabular}{@{}lc@{\hspace{0.05in}}c@{\hspace{0.1in}}r@{\hspace{0.05in}}rc@{\hspace{0.05in}}c@{\hspace{0.045in}}c@{\hspace{0.1in}}c@{\hspace{0.08in}}l@{\hspace{0.05in}}lcl@{}}  %c@{\hspace{0.045in}}
  \hline\hline
ID  & RA (J2000) & Dec. (J2000) &\multicolumn{1}{c}{$X$}&\multicolumn{1}{c}{$Y$}& $V$  & ($B-V$) & 
($V-I$) & [Fe/H] &Type$^\ast$&\multicolumn{1}{c}{bands(exptime)} & PID{ }&{\hspace{0.25in}}Datasets \\ % field ID,&
       &            &             &\multicolumn{1}{c}{arcm} &\multicolumn{1}{c}{arcm} &    &       &     &       &        &       &       &               \\                 % camera$^\dagger$&
 \hline                                                                                                                                                   %      rbc tp caldw
  B004D-V223          & 00 40 26.41 & +41 13 42.7 & --17.82 &  18.98 & 18.81 & 1.18 &      &             &4& {\it F606W} (3250s), {\it F435W} (7260s) & 10407& J96Q07010, J96Q07020 \\  % P1 & 4 4 4 4     B004D-V223$^3$         
&  &  & & & &  &  &  & &  &\\  %  &
  B037-V327               & 00 41 35.00 & +41 14 54.9 &  --8.98 &   9.51 & 16.82 & 2.05 & 2.63 &  --1.07$^a$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z003010, J8Z003020 \\  % F1 & 1 1 1 1     B037-V327              
  B041-G103               & 00 41 40.73 & +41 14 45.8 &  --8.44 &   8.57 & 17.65 & 0.97 & 1.18 &  --1.22$^a$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z019010, J8Z019010 \\  % F2 & 1 1 1 1     B041-G103              
&  &  & & & &  &  &  & &  &  &\\  %  &
  B042-G104               & 00 41 41.69 & +41 07 25.8 & --14.12 &   3.93 & 16.29 & 1.48 & 1.89 &  --1.09$^b$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z060010, J8Z022010 \\  % G1 & 1 1 1 1     B042-G104              
&  &  & & & &  &  &  & &  &\\  %  &
  B057-G118$^\dagger$    & 00 41 52.84 & +40 52 04.6 & --24.96 & --7.15 & 17.64 & 0.69 & 0.99 &  --2.12$^a$ &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q06010, J96Q06020 \\  % C2 & 1 1 1 1     B057-G118$^\dagger$    
  B253                    & 00 41 49.63 & +40 52 59.7 & --24.60 & --6.11 & 18.01 &      & 0.55 &             &6& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q06010, J96Q06020 \\  % C2 & 6 6 6 6     B253                   
  B034D                   & 00 41 50.13 & +40 51 46.7 & --25.51 & --6.93 & 17.50 &      &      &             &6& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q06010, J96Q06020 \\  % C2 & 6 6 6 6     B034D                  
%  SK038A                  & 00 41 50.89 & +40 52 47.1 & --24.62 & --6.42 & 18.20 & 0.87 & 0.98 &             &1& F606W(2110s),F435W(2672s)    & 10407& J96Q06010,J96Q06020 \\  % C2 & 1 0 1       SK038A                 
  B522-SK038A             & 00 41 50.94 & +40 52 48.3 & --24.60 & --6.42 & 17.85 &      &      &             &1(2)& {\it F606W} (2110s), {\it F435W} (2672s) & 10407& J96Q06010, J96Q06020  \\  % C2 & 2 0 2 1     B522                                       
  SK102C                  & 00 41 55.92 & +40 50 19.7 & --25.98 & --8.68 & 15.22 & 0.76 & 0.85 &             &6(2)& {\it F606W} (2110s), {\it F435W} (2672s) & 10407& J96Q06010, J96Q06020  \\  % C2 & 2 0 6 6     SK102C                 
  B521-SK034A             & 00 41 41.67 & +40 52 01.4 & --26.29 & --5.51 &       &      &      &             &1(2)$^\ddagger$& {\it F606W} (2110s), {\it F435W} (2672s) & 10407& J96Q06010, J96Q06020  \\  % C1 & 2 0 2 1y    B521                    
%  SK034A                  & 00 41 41.71 & +40 52 00.9 & --26.30 & --5.53 & 18.89 & 0.57 & 0.65 &             &1& F606W(2110s),F435W(2672s)    & 10407& J96Q06010,J96Q06020  \\  % C1 & 1 0 1       SK034A                 
  B458-D049$^\dagger$    & 00 41 44.61 & +40 51 22.3 & --26.47 & --6.35 & 17.84 & 0.49 & 0.57 &  --1.18$^a$ &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q06010, J96Q06020  \\  % C1 & 1 1 1 1y    B458-D049$^\dagger$         
  B049-G112$^\dagger$    & 00 41 45.60 & +40 49 53.7 & --27.52 & --7.41 & 17.56 & 0.52 & 0.69 &  --2.14$^a$ &1& {\it F606W} (2110s), {\it F435W}(2672s)    & 10407& J96Q06010, J96Q06020  \\  % C1 & 1 1 1 1y    B049-G112$^\dagger$         
  SK036A                  & 00 41 47.34 & +40 51 07.5 & --26.35 & --6.91 & 19.43 & 1.01 & 1.13 &             &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q06010, J96Q06020  \\  % C1 & 1 0 1 1     SK036A                 
&  &  & & & &  &  &  & &  &  &\\  %  &
  B063-G124               & 00 42 00.80 & +41 29 09.5 &    5.24 &  14.43 & 15.66 & 1.21 & 1.58 &  --0.87$^b$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z008010, J8Z024010  \\  % H1 & 1 1 1 1     B063-G124 %ici             
  B061-G122               & 00 42 00.20 & +41 29 35.5 &    5.51 &  14.79 & 16.61 & 1.12 & 1.49 &  --0.79$^b$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z008010, J8Z024010  \\  % H1 & 1 1 1 1     B061-G122              
  G137                    & 00 42 09.43 & +41 28 30.4 &    5.71 &  12.76 & 17.81 & --0.02 &    &             &5& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z008010, J8Z024010  \\  % H1 & 5 5 5 5     G137                   
&  &  & & & &  &  &  & &  &  &\\  %  &
  SK107C              & 00 42 14.18 & +41 34 26.3 &   10.94 &  15.70 & 19.65 & 0.80 & 0.89 &             &6(2)& {\it F606W} (2370s), {\it F814W}(2370s) & 10260& J8Z007010, J8Z023010  \\  % I2 & 2 0 6 6     SK107C$^2$              
  B515$^{1}$                & 00 42 28.05 & +41 33 24.5 &   11.72 &  13.02 & 18.67$^{2}$ &      &      &         &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z007010, J8Z023010  \\  % I2 & 1 1 1 1     B515$^2$               
  B056D               & 00 42 28.45 & +41 34 27.2 &   12.59 &  13.60 & 18.70 &      &      &             &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z007010, J8Z023010  \\  % I2 & 1 1 1 1     B056D$^2$              
  B102                & 00 42 29.85 & +41 34 18.2 &   12.64 &  13.30 & 16.58 & 0.62 & 0.95 &             &7& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z007010, J8Z023010  \\  % I2 & 7 7 7 6     B102$^2$               
&  &  & & & &  &  &  & &  &  &\\  %  &
  B082-G144               & 00 42 15.79 & +41 01 14.3 & --15.06 & --4.94 & 15.54 & 1.56 & 1.91 &  --0.86$^b$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z004010, J8Z020010  \\  % L1 & 1 1 1 1     B082-G144              
  SK094B                  & 00 42 07.81 & +41 01 10.0 & --16.05 & --3.80 & 18.13 & 1.11 & 1.24 &  --0.86$^b$ &4(2)& {\it F606W} (2370s), {\it F814W} (2370s) & 10260& J8Z004010, J8Z020010  \\  % L1 & 2 0 4       SK094B                 
  B090$^\dagger$         & 00 42 21.12 & +41 02 57.3 & --13.09 & --4.68 & 18.80 &      & 1.64 &  --1.39$^a$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z004010, J8Z020010  \\  % L2 & 1 1 1 1     B090$^\dagger$         
&  &  & & & &  &  &  & &  &  &\\  %  &
  B094-G156               & 00 42 25.01 & +40 57 17.2 & --17.11 & --8.74 & 15.55 & 0.97 & 1.26 &  --0.41$^b$ &1& {\it F555W} (413s), {\it F814W} (502s)     & 10273& J92GB9BRQ, J92GB9BPQ \\  % N2 & 1 1 1 1     B094-G156              
  SK048A                  & 00 42 17.59 & +40 55 15.3 & --19.58 & --8.89 & 18.49 & 0.65 & 0.74 &             &1& {\it F555W} (413s), {\it F814W} (502s)     & 10273& J92GB9BRQ, J92GB9BPQ  \\  % N1 & 1 0 1       SK048A                 
&  &  & & & &  &  &  & &  &  &\\  %  &
  B130-G188               & 00 42 48.91 & +41 29 52.9 &   11.35 &   7.77 & 16.93 & 1.15 & 1.41 &  --1.28$^a$ &1& {\it F555W} (413s), {\it F814W} (502s)     & 10273& J92GB6ZLQ, J92GB6ZNQ  \\  % O1 & 1 1 1 1     B130-G188              
  B072D                   & 00 42 45.78 & +41 27 26.9 &    9.07 &   6.74 & 18.50 &      &      &  --1.28$^a$ &3(4)$^3$ & {\it F555W} (413s), {\it F814W} (502s)  & 10273& J92GB6ZLQ, J92GB6ZNQ  \\  % O2 & 4 4 ? 1     B072D                  
&  &  & & & &  &  &  & &  &  &\\  %  &
  B151-G205               & 00 43 09.64 & +41 21 32.6 &    7.17 & --0.43 & 14.83 & 1.23 & 1.45 &  --0.75$^b$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z005010, J8Z021010  \\  % M1 & 1 1 1 1     B151-G205              
  B147-G199               & 00 43 03.31 & +41 21 21.5 &    6.30 &   0.39 & 15.80 & 0.84 & 1.27 &  --0.24$^b$ &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z005010, J8Z021010  \\  % M1 & 1 1 1 1     B147-G199              
  SK077D                  & 00 43 00.52 & +41 23 37.6 &    7.76 &   2.20 & 17.66 & 0.41 & 0.48 &             &6& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z005010, J8Z021010  \\  % M2 & 6 0 6       SK077D                 
  SK078D                  & 00 43 00.86 & +41 22 52.3 &    7.20 &   1.69 & 18.13 & 0.69 & 0.78 &             &6& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z005010, J8Z021010  \\  % M2 & 6 0 6       SK078D                 
  SK079D                  & 00 43 06.04 & +41 22 28.7 &    7.49 &   0.68 & 18.67 & 1.43 & 1.62 &             &6& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z005010, J8Z021010  \\  % M2 & 6 0 6       SK079D                 
  SK120B                  & 00 43 05.97 & +41 23 08.1 &    8.00 &   1.10 & 19.33 & 0.82 & 0.92 &             &6(2)& {\it F606W} (2370s), {\it F814W} (2370s) & 10260& J8Z005010, J8Z021010  \\  % M2 & 2 0 6 6     SK120B                 
  B162-G216               & 00 43 16.42 & +41 24 04.2 &    9.95 &   0.13 & 17.48 & 1.05 & 1.34 &             &1& {\it F606W} (2370s), {\it F814W} (2370s)    & 10260& J8Z005010, J8Z021010  \\  % M2 & 1 1 1 1     B162-G216              
&  &  & & & &  &  &  & &  &  &\\  %  &
  B171-G222               & 00 43 25.67 & +41 15 37.4 &   4.37  & --6.45 & 15.28 & 0.99 & 1.58 &  --0.48$^b$ &1& {\it F606W} (3396s), {\it F435W} (4476s)    & 10407& J96Q03010, J96Q03020  \\  % A1 & 1 1 1 1     B171-G222              
  B169                    & 00 43 23.06 & +41 15 25.5 &   3.91  & --6.19 & 17.08 & 1.23 & 1.31 &             &1& {\it F606W} (3396s), {\it F435W} (4476s)    & 10407& J96Q03010, J96Q03020  \\  % A1 & 1 1 1 1     B169                   
  SK083D                  & 00 43 28.60 & +41 14 36.7 &   3.92  & --7.51 & 14.64 & 1.05 & 1.17 &             &6& {\it F606W} (3396s), {\it F435W} (4476s)    & 10407& J96Q03010, J96Q03020  \\  % A1 & 6 0 6       SK083D                 
  B175                    & 00 43 30.18 & +41 14 36.4 &   4.09  & --7.74 & 16.80 & 0.80 &      &             &6& {\it F606W} (3396s), {\it F435W} (4476s)    & 10407& J96Q03010, J96Q03020  \\  % A1 & 6 6 6 6     B175                   
  B185-G235               & 00 43 37.41 & +41 14 43.3 &   5.02  & --8.74 & 15.54 & 0.94 & 1.18 &  --1.03$^b$ &1& {\it F606W} (3396s), {\it F435W} (4476s)    & 10407& J96Q03010, J96Q03020  \\  % A2 & 1 1 1 1     B185-G235              
&  &  & & & &  &  &  & &  &  &\\  %  &
  B206-G257               & 00 43 58.70 & +41 30 18.0 &   19.74 & --2.25 & 15.06 & 0.80 & 1.03 &  --1.45$^b$ &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q05010, J96Q05020  \\  % D1 & 1 1 1 1     B206-G257                   
  B198-G249               & 00 43 50.07 & +41 31 53.1 &   19.99 & --0.00 & 17.55 & 0.60 & 1.11 &  --1.13$^a$ &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q05010, J96Q05020  \\  % D1 & 1 1 1 1     B198-G249              
  B203-G252               & 00 43 56.00 & +41 32 36.0 &   21.23 & --0.43 & 16.68 & 0.93 & 1.20 &  --0.90$^a$ &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q05010, J96Q05020  \\  % D2 & 1 1 1 1     B203-G252              
  B213-G264               & 00 44 03.62 & +41 30 38.9 &   20.58 & --2.76 & 16.78 & 1.05 & 1.29 &  --0.99$^b$ &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q05010, J96Q05020  \\  % D2 & 1 1 1 1     B213-G264              
  SK079A                  & 00 44 04.58 & +41 32 09.3 &   21.88 & --1.97 & 18.63 & 1.11 & 1.23 &             &6(1)& {\it F606W} (2110s), {\it F435W} (2672s) & 10407& J96Q05010, J96Q05020  \\  % D2 & 1 0 6 6     SK079A                 
  B215-G266               & 00 44 06.44 & +41 31 43.9 &   21.76 & --2.51 & 17.13 & 1.02 & 1.20 &             &1& {\it F606W} (2110s), {\it F435W} (2672s)    & 10407& J96Q05010, J96Q05020  \\  % D2 & 1 1 1 1     B215-G266              
&  &  & & & &  &  &  & &  &  &\\  %  &
  M039$^4$                & 00 44 31.30 & +41 30 04.6 &   23.34 & --7.18 & 18.94 & 1.11 &--0.53&             &1(2)$^\ddagger$& {\it F606W} (1860s), {\it F435W}(2910s) & 10407& J96Q02010, J96Q02020  \\  % Q2 & 2 0 2 1y    M039$^4$                 
&  &  & & & &  &  &  & &  &  &\\  %  &
  B234-G290               & 00 44 46.50 & +41 29 18.3 &  24.50  & --9.90 & 16.78 & 1.00 & 1.18 &  --0.95$^a$ &1& {\it F606W} (3315s), {\it F435W} (4560s)    & 10407& J96Q04010, J96Q04020   \\  % B1 & 1 1 1 1     B234-G290              
  M047                    & 00 44 37.85 & +41 28 52.1 &  23.16  & --8.90 & 18.84 &      &  1.2 &             &2$^5$ & {\it F435W} (4560s), {\it F606W} (3315s) & 10407& J96Q04010, J96Q04020  \\  % B2 & 2 0 2 1     M047                   
  B231-G285               & 00 44 38.61 & +41 27 46.8 &  22.39  & --9.68 & 17.27 & 0.84 & 1.14 &  --1.49$^a$ &1& {\it F435W} (4560s), {\it F606W} (3315s)    & 10407& J96Q04010, J96Q04020  \\  % B2 & 1 1 1 1     B231-G285              
  M050                    & 00 44 40.59 & +41 30 06.0 &  24.44  & --8.53 & 18.71 &      & 0.40 &             &1(2)$^\ddagger$& {\it F606W} (3315s), {\it F435W} (4560s) & 10407& J96Q04010, J96Q04020  \\  % B1 & 2 0 2 1y    M050                   
&  &  & & & &  &  &  & &  &  &\\  %  &
  B367-G292$^{\dagger}$ & 00 44 47.18 & +42 05 31.9 &  53.00 &  12.48 & 18.45 & 0.32 & 1.30 &  --2.32$^a$ &1& {\it F606W} (1850s), {\it F435W} (2920s)    & 10407& J96Q01010, J96Q01020   \\  % E2 & 1 1 1 1y     B367-G292$^{1,\dagger}$
  SK181B              & 00 44 48.64 & +42 06 08.1 &   53.64 &  12.64 & 19.18 & 1.28 & 1.46 &             &6(2)& {\it F606W} (1850s), {\it F435W} (2920s) & 10407& J96Q01010, J96Q01020  \\  % E2 & 2 0 6 6     SK181B$^1$             
\\
 \hline 
                                                                              \end{tabular} 
			                  }
\par
			                  \smallskip
\par
 {{All of them were observed with WFC@ACS on HST. ID, coordinates and photometry are from Galleti et~al. (\cite{Galleti04}). We note the double identifications B521~=~SK034A and B522~=~SK038A.
$^a$: Perrett et~al. (\cite{Perrett02});  $^b$: Huchra et~al.  (\cite{Huchra91});  
$^{\dagger}$: BLCC, Fusi Pecci et~al. (\cite{FusiPecci05})}
}
 {{$^{\ast}$: classification, coded as follows: 1- confirmed cluster; 2- gc candidate; 3- controversial object 4- galaxy; 
5- HII~region; 6- star; 7- asterism;
 ~~$^\ddagger$: young cluster (from this paper and/or Caldwell et~al. (\cite{Caldwell09})). In parentheses is enclosed the previous RBCv3.5 value. }
}
{{$^1$: identified as KHM31-409 in Krienke \& Hodge (\cite{Krienke08}), Table~4; 
 $^2$: $V$~mag from 
Krienke \& Hodge (\cite{Krienke08}), Table~4; 
$^3$: B072D, that was originally classified as a galaxy by Huxor et~al. (\cite{Huxor08}), looks like a cluster, as noted also 
by Caldwell et~al. (\cite{Caldwell09}). Radial velocity is necessary  in our view to yield its firm confirmation; 
 $^4$:~identified as KHM31-516 in Krienke \& Hodge (\cite{Krienke08}), 
Table~4;  $^5$: classified as globular cluster by Caldwell et~al. (\cite{Caldwell09}).} 
}
\vspace*{2mm}
\end{table}